Author's response to reviews

Title: Epidemiological Trends in Nosocomial Candidemia in Intensive Care Unit (ICU) in Italy

Authors:

Matteo Bassetti (mattba@tin.it)
Elda Righi (mattba@tin.it)
Alessandro Costa (mattba@tin.it)
Roberta Fasce (mattba@tin.it)
Maria Pia Molinari (mattba@tin.it)
Franco Bobbio Pallavicini (mattba@tin.it)
Claudio Viscoli (mattba@tin.it)
Raffaella Rosso (mattba@tin.it)

Version: 5 Date: 9 January 2006

Author's response to reviews:

Dear Editor,

Thank you for your message of 6/1/06 and closed comments. However the type of study (retrospective) do not need the approval by the ethical review board. We added in the text that all the patients consented to participation in the study and publication of the results. We added also the section: competing interests, author contributions and acknowledgements. The changes in the revised manuscript are in red.

We hope that the revised version of the manuscript will be now suitable for the publication in the BMC Infectious Disease.

Thanking you in advance, I look forward for hearing from you.

Matteo Bassetti